Author's response to reviews

Title: Evaluation of the wound healing property of Commiphora guidottii Chiov. ex. Guid.

Authors:

Michael Gebrehiwot (mgebrehiwot@gmail.com)
Kaleab Asres (kaleab.asres@aau.edu.et)
Daniel Bisrat (daniel.bisrat@aau.edu.et)
Avijit Mazumder (avijitmazum@yahoo.com)
Peter Lindemann (peter.lindemann@pharmazie.uni-halle.de)
Franz Bucar (franz.bucar@uni-graz.at)

Version: 2 Date: 11 December 2014

Author's response to reviews:

1. Please revise your manuscript to include line and page numbers.
   Manuscript now contains line and page numbers

2. At present, we do not feel that there is sufficient evidence presented in your Background section to justify the testing of Commiphora guidottii Chiov. ex. Guid. in an animal model of wound healing. We would therefore ask you to expand this section to include as much referenced evidence as possible to explain why you would expect this treatment to have an effect in this model.

   The background has been expanded by including previous in vitro, isolated organ and animal work done on the plant and its constituents.